RICH1 Phone Meeting 3rd August 07

Present:

Bristol: Nick Brook, Richard Head

Oxford: Brian Ottewell, Neville Harnew

Imperial College: Bill Cameron, Richard Plackett, Dave Websdale
CERN: Clara Matteuzzi, Jon Imong, Olav Ullaland, Werner Witzeling

1) Gas enclosure

1.1) Status

Erich has repaired the dry air supply and installed a line to RICH1. There is one volume change of dry air every three hours.

The bosses on the inside of the GE doors that clashed with the spherical mirrors have been machined off, exposing the holes that have to be plugged.  Tim Barrass (Bristol) has produced some plugs and Richard Head will produce some seals to accompany these. (Action) Bill and or Dave will provide measurements to Richard for this.

The spherical mirrors have been protected from accident by some metal half doors that have been installed.

The Aerogel mounts were not left in position; they will be mounted in the first week in September. Milano would also like to study the possibility of making a tool to help install the boxes. There is a time cut off in September with the magnet test scheduled for the last week in September.  The RICH box has to be closed at least two weeks before this to allow the magnetic doors to be put in position and a proper gas leak test to be carried out..
1.2) Temperature Sensors

No recent change – Installed but require patch panel and a different connector.

2) Magnet Tests

To close the GE for the magnet test clearances need to be checked at each step to protect the mirrors, GE doors and the cables.

When the magnet test takes place we would like sensors on both halves of the shielding to detect any movement. This is especially valuable for the bottom shielding as it has not been tested with the current floor clamps.

We need to make sure no equipment is left in the RICH1 area when the magnet is turned on.

3) Mirrors

3.1) Installation

The mirror installation was completed last week

Monday – The mirrors were transported and the GE was cleaned.

Tuesday – The spherical mirrors were installed, with a provisional modification to the rail clamps (the definitive clamps have since been fit at the central V-blocks)
Wednesday – The flat mirrors were installed

Thursday – The survey was performed.

The results of the survey arrived this morning. The upper flat mirrors are at an angle of 248.7mrad the lower flat mirrors are at an angle of 253.2mrad (nominal is 250mrad).

The spherical mirrors are positioned correctly in x but are 1mm too low in y. This offset was expected and can be adjusted with shims.
The centres of curvature of the spherical mirrors are 10mm too low. This is due to a known displacement of the lower GE shelf. Whilst these inaccuracies are within our limits of tolerance it would be preferable to make adjustments to correct them.  This should be done before the 1st week in September and we must liaise with the survey team to check after the adjustment.  

(Action) Jon will analyse the survey data and determine the corrections that have to be made.

4) HPD assembly

4.1 Column Dressing

We have 14 RICH1 columns ready to be installed in the HPD boxes.

The rate of failure of the regulator temperature sensors has improved from the RICH2 column dressing to of the order one failure after the plate has been attached.
The spare column mechanics are at CERN and the new cups have been delivered.

The spare columns will be dressed in September.

4.2 Upper Rail System

The final parts we are waiting for will be delivered in time to be shipped to CERN on Sunday.

The Chariot went to Oxford last Friday (27/7/07)
The external rail system was set up. The plinth on the chariot clashed with a cross member on the external rails.

The trial installation at CERN next week can proceed with the plinth removed.

Before any corrective action can be taken we must wait for Trevor and Rohan to return from vacation. The member doesn’t seem to be load bearing.
4.3 Schedule for Mounting and Testing Upper Rails

The parts will be delivered to IP8 on Monday morning (6/8/07).

The internal rails will be installed and their fit checked.

The external rails will be installed and their fit checked.

The HPD chariot will be installed without the plinth to make sure it runs on the rails into its final position.

The external rails will be load tested.

Two HPD columns will be installed in the frame to check close packing in the HPD box.

The HPD frame and some parts of the external rails will then return to the UK for modification.

The main shaft crane will be out of action, however we should be able to use the passenger lifts to transport the components.

4.4 Lower Rail system

The model for the bottom rails is progressing well and Brian is adding details. 

It is hoped that the model will be completed by next Friday.

The model will then be checked against the designs by Trevor for compatibility.

4.5 Status of Mechanics

Trevor is currently focussing on the design for the RICH1 cooling manifold. This is more complicated than the RICH2 version due to the horizontal orientation of the columns.

4.6 HPD Cabling
There has been no change to the cabling since the last meeting.

4.7 Light Leak Detector

A decision still has to be made on the choice of PMT system

4.8 Photon Funnel

Measurements will be made to make best use of the space available between the GE and shield, so we can provide as much space as possible for the MDM.

5 HPD cooling

See 4.5

6 LAMS

All the components have been received at Imperial.

A focussed spot from a laser-injected fibre has been recorded on the CID camera.  

One problem seems to be that the direction of the focussing unit is unstable during adjustment.

7 MDM

Four stages have been received but the drivers have yet to arrive.

The stages are as expected. No nasty surprises.

Currently the light source and the collimators are being optimised.

We still require photon funnel(PF)  mechanical and cabling requirements to begin PF manufacture

Resuming regular meetings on the MDM would be very helpful.

It would be helpful for Ray to provide a paragraph report for the minutes of these meetings (action Dave).
8 Quartz Windows

The quartz windows are ready. The schedule for gluing is driven by the PF manufacture.

9 Aerogel

The company providing the aerogel has had a production problem involving its autoclave.

For every 1 tile produced 8 are rejected.

The have agreed to provide only 4 litres of aerogel. This should be sufficient. However, this could mean we have no manufacturer for the future for contingencies.

The autoclave problem has occurred and been rectified before.

We will make it clear that we would be very interested in resuming production when they have the capacity back.

